%%%=============================================%%%
%   Example for preparing a paper to EuroSim'04   %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\documentclass[11pt]{eurosim}
\ifx\pdfoutput\undefined
    \usepackage[dvips]{graphicx}
    \DeclareGraphicsExtensions{.ps,.eps}
\else
    \usepackage[pdftex]{graphicx}
    \DeclareGraphicsExtensions{.png,.jpg}
\fi

\oddsidemargin -0.03 in
\evensidemargin -0.03 in
\topmargin=-1,3cm
\textheight=24,5cm
\textwidth=16cm

\title{AUTHOR'S INSTRUCTIONS\\
FOR THE PREPARATION OF PAPERS FOR\\ THE EUROSIM 2004 CONGRESS}

\author{First Author$^1$, Second Author$^2$, Third Author$^1$, Fourth Author$^1$\\[2ex]
%
\normalsize{$^1$First, Third and Fourth Author's Affilation,
Address, Country}\\
\normalsize{ \{first,third,fourth\}@affilation1.eu}\\
\normalsize{ http://www.first.com}\\
\normalsize{ $^2$Second Author's Affilation, Address, Country}\\
\normalsize{ second@affilation2.eu}}

\pagestyle{empty}
\date{}

\begin{document}
\begin{thebibliography}{99}
\bibitem{Abb79} M.B. Abbott (1979). {\em Computational Hydraulics.} London: Pitman.
\bibitem{Ber94MS} L. Bergamaschi, S.~Mantica and F.~Saleri (1994). Mixed finite element approximation of {D}arcy's law in porous media. Report CRS4 AppMath-94-20, CRS4, Cagliari, Italy.
\bibitem{Bud80U} W.P. Budgell and T.E. Unny (1980). A stochastic-deterministic model for predicting tides in branched estuaries. In {\em Proceedings of the 3th Int. Symp. on Stoch. Hydr.} Tokyo.
\bibitem{Eve94} E. Evensen, (1994). Sequential data assimilation with a nonlinear {QG} model using {M}onte {C}arlo methods to forecast error statistics. {\em J. Geophys. Res.\/}~{\bf 99}, 10143--10162.
\bibitem{Hou98M} P.L. Houtekamer and A.L. Mitchel (1998). Data assimilation using ensemble kalman filter technique. {\em Monthly Weather Review\/}~{\bf 126}, 796--811.
\end{thebibliography}

\maketitle
\thispagestyle{empty}


%%------------------ Abstract Part ----------------
\noindent
{\large \bf ABSTRACT}
\\[2ex]
%
The abstract should summarize the contents of the paper in
approximately 100 words. Please do not enter formulas or
references in the abstract.
\\[4ex]
%
{\large \bf KEYWORDS}
\\[2ex]
%
Keyword, Keyword, Keyword, Keyword, Keyword.
%

%--------------------- Paper Introduction -------------
\section{INTRODUCTION}
%
The EUROSIM 2004 Congress will be organized by the French Speaking
Simulation Society (FRANCOSIM) on behalf of the Federation of
European Simulation Societies (EUROSIM). It will take place in
September 2004 at ESIEE in Marne la Valle\'e, France.

%--------------------- First Section -------------------
\section{GENERAL GUIDELINES}
\label{sec1}
%
Please write your paper in English, on A4 paper format with 2,5 cm
margins (left, right, top and bottom). Use Times New Roman font,
and, with the exception of the title, authors and affiliations,
and headings, the text should be fully-justified (left and right).
The paper should not exceed six pages. Do not insert page numbers
in the paper.\\

Please write the references in the form of
\cite{Abb79,Ber94MS,Bud80U}, \cite{Eve94}, \cite{Hou98M}. text
text text text text text text text text.\\

Please write the equation as follows:
\begin{equation}
C=A+B
\label{eq:1}
\end{equation}

Each equation should has a number as given in (\ref{eq:1})

%--------------------- Second Section -------------------
\section{SPECIFIC GUIDELINES}
\label{sec2}
%
\subsection{Recommended Order of Contents}
\label{subsec2.1}
%
Text of subsection.
\begin{equation}
X=X+1
\label{eq:2.1}
\end{equation}
text text text .
%
\subsubsection{Tables}
\label{subsubsec2.1.1}
%
Text of subsubsection. In the Table~\ref{tab:1}.......
%
\begin{table}[htb]
\caption{Table}
\label{tab:1}
\begin{center}
%\setlength{\extrarowheight}{4pt}
\begin{tabular}{lll}
\hline
\hline
Heading level &Style &Font size\\
\hline
{\Large \bf TITLE} & bold, fully, capitalized &14-point\\
{\large \bf ABSTRACT} & bold, fully, capitalized &12 -points\\
\hline
\hline
\end{tabular}
\end{center}
\end{table}
text text text .

Text text text text.\\
%
\subsubsection{Figure}
\label{subsubsec2.1.2}
%
Text of subsubsection. In the Figure~\ref{fig1} ....
%
\begin{figure}[htb]
\centerline{\includegraphics{fig1}}
\caption{Text of fig-caption}
\label{fig1}
\end{figure}
text text text .


%--------------------- Acknowledgement Section -------------------
\section*{ACKNOWLEDGEMENTS}
%
Text of acknowledgements.


%--------------------- Appendix Section -------------------
\appendix
\section{Proofs}
\label{app1}
\setcounter{equation}{0}
%
Text of appendix A.
\begin{equation}
Z=\sum_p \sum_{i \in TC_p} C_{ip} \label{eq-app:1}
\end{equation}
text text text .
%

%--------------------
\section{Results}
\label{app2}
\setcounter{equation}{0}
%
Text of appendix B.
\begin{equation}
Z=\sum_p \sum_{i \in TC_p} C_{ip} \label{eq-app:2}
\end{equation}
text text text .

% ------------------------ Bibliography ----------------------

%
\end{document}
%
